Author's response to reviews

Title: Association between congenital toxoplasmosis and parent-reported developmental outcomes, concerns, and impairments, in 3 year old children.

Authors:

Katherine Freeman (kfreeman@montefiore.org)
Alison Salt (a.salt@ich.ucl.ac.uk)
Andrea Prusa (arprus@hotmail.com)
Gunilla Malm (gunilla.malm@hs.se)
Nicole Ferret (ferret.n@chu-nice.fr)
Wilma Buffalono (wilma@cds.unina.it)
Dorthe Schmidt (DRS@ssi.dk)
Hooi Kuan Tan (h.tan@ich.ucl.ac.uk)
Ruth E Gilbert (r.gilbert@ich.ucl.ac.uk)
European Multicentre Study on Congenital Toxoplasmosis (emscot) (r.gilbert@ich.ucl.ac.uk)

Version: 6 Date: 5 July 2005

Author's response to reviews:

Table 1 had been split into 2 parts. The table has now been merged.